Author’s response to reviews

Title: ASSOCIATION BETWEEN EARLY CHILDHOOD CARIES AND MALNUTRITION IN A SUB-URBAN POPULATION IN NIGERIA

Authors:

Morenike Oluwatoyin Folayan (toyinukpong@yahoo.co.uk)

Olujide Arije (olujide_arije@yahoo.com)

Maha El Tantawi (maha_tantawy@hotmail.com)

Kikelomo Adebunke Kolawole (kikelomokolawole@gmail.com)

Mary Obiyan (mobiyan@cartafrica.org)

Olaniyi Arowolo (arotayniy@yahoo.com)

Elizabeth Oziegbe (elioziegbe@yahoo.com)

Version: 4 Date: 28 Oct 2019

Author’s response to reviews:

-- If consent to participate statements are already present within the main text/Methods section, please also copy/place it under the 'Ethics approval and consent to participate' sub-section. DONE

-- We note that you have included a ‘Consent for publication’ section in the Declarations. Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please state “Not Applicable” in this section. done